Author’s response to reviews

Title: The Scandinavian Multicenter Hemodynamic Evaluation of the SJM Regent Aortic Valve

Authors:

Jon Offstad (jon.offstad@rikshospitalet.no)
Kai Andersen (Kai.Andersen@Rikshospitalet.no)
Per Paulsson (per.paulsson@skane.se)
Jesper Andreasson (jesper.andreasson@skane.se)
Ulf W Kjellman (ulf.w.kjellman@vgregion.se)
Oluf Lundblad (olof.lundblad@orebro.se)
Karl Gunnar Engstøm (gunnar.engstrom@vll.se)
Rune Haaverstad (rhaavers@online.no)
Jan L Svennevig (j.l.svennevig@medisin.uio.no)

Version: 4 Date: 15 December 2011

Author’s response to reviews:

Tables are now sited in the correct order.
Tables have been divided into rows and columns. Unfortunately the Hospital is still using an old version of Word without this new option, so I have done my best on my own PC. If the format is still wrong, I may need your help.
I don’t know what title in "sentence case" means, I assume it means ordinary text?
Please fint enclose a revised manuscript.
Best regards
Jan L. Svennevig